Author's response to reviews

Title: Effectiveness of Chinese herbal medicine for impaired glucose tolerance

Authors:

SUZANNE J GRANT (S.GRANT@UWS.EDU.AU)
DENNIS H CHANG (D.CHANG@UWS.EDU.AU)
JIAXUN LIU (liujx0324@sina.com)
VINCENT WANG (Vincent.Wong@sswhs.nsw.gov.au)
HOSEN KIAT (HOSEN.KIAT@CHI.ORG.AU)
ALAN BENSOUSSAN (A.BENSOUSSAN@UWS.EDU.AU)

Version: 2 Date: 24 June 2012

Author's response to reviews:

Dear Editors,

As requested, we have made the following changes to the manuscript:

1) We have incorporated the trial registration number in the abstract
2) We have provided a CONSORT checklist (see attached)
3) We have provided a 'Competing Interests' section

Many thanks for considering this manuscript.

Yours sincerely

Dennis Chang